 Vc/6-Sep-04
Week37 starting 6Sept  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 30 aug04  ~14.20 hrs 

	TBA1
	MBBL1126
	Prep 5 Pumping + Cool Down
	302.5

	TBA2
	MBAL2047
	PT 12 Warm Up
	4.6

	TBB1
	SSS035
	FS21 MQ strength@760
	1.9

	TBB2
	_
	_
	300.3

	TBC1
	MBAL3202
	ICS 4 Final connection
	298.6

	TBC2
	MBBL1099
	ICS 4 Final connection
	299.2

	TBD1
	SSS043
	PT 1 IAP @ cold
	1.91

	TBD2
	SSS038
	Finished Ready to go Final scan
	299.7

	TBE1
	MBAL3159
	Prep 5 Pumping + Cool Down
	270.1

	TBE2
	MBBL3195
	Prep 5 Pumping + Cool Down
	92.8

	TBF1
	MBAR1133
	PT 12 Warm Up
	101.5

	TBF2
	MBAL3146
	PT 6.1 Training 1
	26.3


9/6/2004 2:20:30 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Marco; MTM support on call :  (16-3602).

11 Magnets [ 9  dipoles( 3 with MM)+ 2 SSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· ( SSS work : New issues like Wire Breakages

	Magnets under Test from 29-08-2004 to 04-09-2004   LAST WEEK

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAL3152 
	[image: image1.png]



	30-08-2004 
	23 h.
	41 h.
	18 h.
	26 h.
	6 h.
	04-09-2004 
	120 hours
	
	

	2
	TBA1
	MBAL3199 
	[image: image2.png]



	23-08-2004 
	28 h.
	53 h.
	34 h.
	28 h.
	7 h.
	30-08-2004 
	168 hours
	
	

	3
	TBA2
	MBBL2093 
	[image: image3.png]



	26-08-2004 
	30 h.
	27 h.
	27 h.
	26 h.
	6 h.
	31-08-2004 
	120 hours
	
	

	4
	TBB1
	SSS018 
	[image: image4.png]



	17-08-2004 
	81 h.
	45 h.
	85 h.
	7 h.
	73 h.
	30-08-2004 
	312 hours
	
	

	5
	TBB2
	MBBL3196 
	[image: image5.png]



	24-08-2004 
	38 h.
	29 h.
	50 h.
	19 h.
	7 h.
	30-08-2004 
	144 hours
	
	

	6
	TBC1
	MBAL3145 
	[image: image6.png]



	25-08-2004 
	33 h.
	65 h.
	68 h.
	25 h.
	0 h.
	02-09-2004 
	192 hours
	
	

	7
	TBD1
	SSS044 
	[image: image7.png]



	18-08-2004 
	152 h.
	47 h.
	83 h.
	12 h.
	23 h.
	31-08-2004 
	312 hours
	
	

	8
	TBE1
	MBAL3169 
	[image: image8.png]



	28-08-2004 
	26 h.
	31 h.
	39 h.
	23 h.
	3 h.
	02-09-2004 
	120 hours
	
	

	9
	TBE2
	MBBR1036 
	[image: image9.png]



	26-08-2004 
	46 h.
	29 h.
	57 h.
	14 h.
	6 h.
	01-09-2004 
	144 hours
	
	

	10
	TBF1
	MBAL2057 
	[image: image10.png]



	22-08-2004 
	30 h.
	60 h.
	42 h.
	35 h.
	13 h.
	29-08-2004 
	168 hours
	
	

	11
	TBF2
	MBAR1140 
	[image: image11.png]



	24-08-2004 
	26 h.
	44 h.
	64 h.
	17 h.
	5 h.
	31-08-2004 
	168 hours
	
	


Quench Information last week
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3152
	12647 A.
(D1 L)
	12850 A.
(No Data)
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3199
	11419 A.
(D1-L)
	12653 A.
(D2-U)
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL2093
	11123 A.
(D1-U)
	12358 A.
(D2-L)
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS018
	11295 A.
(QI_12)
	12331 A.
(QE_12)
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL3196
	9805 A.
(D2_L)
	11857 A.
(D2_L)
	12850 A.
[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL3145
	10959 A.
(D1_U_S01)
	11571 A.
(D2-U-S01)
	11781 A.
(D2_U_S01)
	11996 A.
(D1_U_S13)
	12184 A.
(D2 U)
	12477 A.
(D1-U-S01)
	12546 A.
(D2-U-S01)
	12715 A.
(D1-U-S01)
	12850 A.
(L0409010)
	[image: image17.png]



	-
	-
	-

	7
	SSS044
	12067.3 A.
(QE-34)
	12856 A.
(No Data)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL3169
	12648 A.
(D1 U)
	12850 A.
(No Data)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR1036
	11825 A.
(D1-L)
	12213 A.
(D1-L)
	11950 A.
(D1_L)
	12170 A.
(D1-L)
	12246 A.
(D1_L)
	12382 A.
(D1_L)
	12432 A.
(D1 L)
	12498 A.
(D1-L)
	12760 A.
(D2-L)
	[image: image20.png]



	-
	-
	-

	10
	MBAL2057
	12315 A.
(D1-U-S01)
	12850 A.
[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR1140
	12465 A.
(D1_U, S1)
	12666 A.
(D1_L-S13)
	12811 A.
(D2-U-S12)
	[image: image22.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


Bench status  and tests for now( Monday 30 Aug/at 17.45 hrs)  & next few days :  see Video Display

General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_31_A
	TBC
	LHC cycle measurements: no reading of the Temperature.
	MTM
	
	Connection still in SSS configuration. Export blocked.

Why not detected in MM0?

	Tuesday_31_A
	TBD2
	Powering corrector circuit CD (MCS) : quench well below 600 A. ie 220 A. Same problem for EF (MQT) : 250 A. 
	ICS?
	
	Investigation by Richard. Everything in the quench detection is OK.

Wit for the disconnection to seen if everything is properly connected.



	Tuesday_31_A 
	TBA2
	HV (WP04) executed by ICS but not available in HVMTF folder.
	ICS
	
	ICS informed : They executed the export later.

	Tuesday_31_N
	TBD2
	SSW at cold : strength measurements 

Message “checking motion status” displayed continuously. 

Test re-launched: but unable to adjust the wire tension.

Test re-launched : wire broke.
	MTM
	
	Richard contacted: he tightened a screw.

	Wednesday_01_A

Wednesday_01_N

Wednesday_01_N


	TBD2
	SSW measurements : SPA abort appeared.

Difficulties to re-start TEMA: power App was hung

SPA again (2 times) during SSW measurements at cold

Wire broke again  during roll angle measurements
	AB-CO
	Power
	Re-start work-station  and re-start TEMA several times.

	Wednesday_01_A


	TBB1
	Power QH OK interlock was not clearing 
	MTM
	
	Richard called 

Origin: Configuration for protection not the same for SSS (TBB) and dipoles (TBB2) with two circuits in instead of 4. So even the PS are charged the PLCSE is not OK when we switch to SSS  because 2 circuits are opened. 

OK when you connect the channels 2 and 3 to dummy resistance.

	Wednesday_01_A


	TBB1
	IAP@warm : polarity on Vtaps inverted.


	MTM
	
	Error in DAQ templates found by Richard.

	Thursday_02_A
	TBC
	PLC fault through the switch is in CC position
	MTM
	
	Solved by Gorgio

	Thursday_02_A
	TBF
	Tcoil  (IAP): wrong values

Problems in coils 0405 –A and coils 1213-E.
	MTM
	
	Richard informed :Tcoil calibration data missing in MTF

Richard changed the setting using E coil for 0405 sector

	Thursday_02_N
	TBD2
	SSW error message (“too many”) after typing the command “menu”
	MTM
	
	Richard found that the “db. seriesRun” file had a wrong variable name.

	Thursday_02_N
	TBD2
	Wire broke during rotwire.
	MTM
	
	OP team waited until the program stopped but the wire still try to stretch the broken wire by turning the motor,

	Friday_03_N

Saturday_04_M
	TBD1
	SSS038: HV test failed orbit dipole vs ground at 1500 V (OK at 300 V).
	MTM
	
	Confirmed by Richard. NC opened.,

	Sunday_05_A
	TBB1
	SPA occurred during XY measurements on SSS035 (2 times).

Then wire broke during XY measurements.
	MTM
	
	

	Sunday_05_A
	TBC2
	IAP @ warm: QH could not be charged. Trigger always present.
	MTM
	
	Richard called.

	Sunday_05_N
	All benches
	General cut of power: (failure in the meyrin site)

Cryo down.
	
	
	Power back at midnight.








